Author's response to reviews

Title: Does treatment of subsyndromal depression improve depression and diabetes related outcomes: protocol for a randomised controlled comparison of psycho-education, physical exercise and treatment as usual

Authors:

Mirjana Pibernik-Okanović (pibernik@idb.hr)
Dea Ajduković (dea.ajdukovic@idb.hr)
Marijana Vučić Lovrenčić (vucic@idb.hr)
Norbert Hermanns (hermanns@diabetes-zentrum.de)

Version: 2 Date: 10 January 2011

Author's response to reviews:

Dear Editors,

Thank you for your e-mail requesting proofing of our manuscript.

In accordance with your recommendations, the title has been changed to: "Does treatment of subsyndromal depression improve depression and diabetes related outcomes: protocol for a randomised controlled comparison of psycho-education, physical exercise and treatment as usual".

The titles and figure numbers have been omitted from the figure files and added to the manuscript after the reference list.

Reference 29 had been cited on page 5, line 7 of the second paragraph, for which reason no additional changes have been made in the current revision.

Corrections of the style of parentheses have been made in lines 1-2 of the first paragraph on page 13.

With kind regards,

Yours sincerely,

Mirjana Pibernik-Okanović